Author's response to reviews

Title: Status on preventing HBV transmission from mother to infant during Pregnancy in China: A retrospective study

Authors:

Guo Yan (swallow315@126.com)
Meng Leping (34763784@qq.com)
Hu Meina (sanliuhu@163.com)
Du Yukai (duyukai100@yahoo.com.cn)

Version: 2 Date: 5 May 2009

Author's response to reviews:

Wuhan, May.10, 2009

RE: Status on preventing HBV transmission from mother to infant during Pregnancy in Whuhan China: A retrospective study, by Guo Yan, Meng Liping, Hu Meina, Du Yukai

Dear BMC Infectious Diseases:
I have used a professional copyediting service to help me copyedit the paper. Some words and sentences had been amended. If it need further revision, please contact with me. Thanks very much.

Guo Yan

Sincerely yours,

Du Yukai